Author's response to reviews

Title: Congenital rubella syndrome in Iran

Authors:

Jila Sadighi (jilasadigh@yahoo.com)
Hassan Eftekhar (info@ihsr.ac.ir)
Kazem Mohammad (info@ihsr.ac.ir)

Version: 6 Date: 31 May 2005

Author's response to reviews:

Dear Iratxe Puebla,

I make the formatting changes listed below:
- removed the repeated title on page 2
- removed unnecessary capitals in the title
- renamed "Competing interests"
- moved apostrophe: Authors’ contributions

Minor Essential Revisions:
1) I search again and found the new reference (MMWR) and added it as reference number 9. Previous number 9 reference was replaced with this one.
2) I explained more about surveillance system in background (page 6).
3) In results, "D" is derived from previous studies (I changed it).
4) I removed "comments for future studies" in discussions, because after rubella vaccination in Iran, epidemiological pattern of disease is changed and we can not do some studies.

Best Regards,

Dr. Jila Sadighi